Author's response to reviews

Title: PARTNERS2: A protocol for the development of core outcome sets for use in effectiveness trials involving people with bipolar and schizophrenia in a community based setting.

Authors:

Thomas Keeley (t.j.h.keeley@bham.ac.uk)
Humera Khan (H.S.Khan@bham.ac.uk)
Vanessa Pinfold (VanessaPinfold@mcpin.org)
Paula Williamson (P.R.Williamson@liv.ac.uk)
Jonathan Mathers (J.M.Mathers@bham.ac.uk)
Linda Davies (Linda.M.Davies@manchester.ac.uk)
Ruth Sayers (RuthSayers@mcpin.org)
Elizabeth England (E.J.England@bham.ac.uk)
Siobhan Reilly (S.Reilly@lancaster.ac.uk)
Richard Byng (Richard.Byng@plymouth.ac.uk)
Linda Gask (Linda.Gask@manchester.ac.uk)
Mike Clark (m.c.clark@lse.ac.uk)
Peter Huxley (p.huxley@bangor.ac.uk)
Peter Lewis (peter.lewis@bsmhft.nhs.uk)
Maximillian Birchwood (m.j.birchwood@warwick.ac.uk)
Melanie Calvert (m.calvert@bham.ac.uk)

Version: 2 Date: 17 November 2014

Author's response to reviews:

Dear Sir/Madam,

Please see details of requested formatting changes and subsequent changes made to the paper.

1. Please move your funding information to an Acknowledgements section before the reference list.

Funding information placed in Acknowledgements section.

2. Please include a statement in your Methods section explaining that you obtained informed consent from each participant.

Statement placed on page 6 para 1.

3. Please include a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. Any experimental research on humans must be in compliance with the Helsinki Declaration. If your study is taking place in multiple sites please ensure you mention all of the ethical bodies that have
approved the study at the different sites.

Statement of ethical approval statement placed on page 5 para 3.

4. Please include a Trial Status section after the Discussion. This should state the status of the trial at the time of manuscript submission. The journal considers study protocol articles for proposed or ongoing trials provided they have not completed patient recruitment at the time of submission.

Statement of study status placed on page 16 para 1.

5. Please include a list of abbreviations used and their meanings, after the Trial Status.

List of abbreviations included on page 16

6. Please include a Competing Interests section after the list of abbreviations. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

Statement included on page 16

7. Please include an Authors’ Contributions section at the end of the manuscript, after the Competing Interests section. Each author needs to be listed individually. We suggest the following kind of format (please use initials to refer to each author’s contribution): “AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.”

Statement included on page 16.

8. Please include an Acknowledgements section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Statement included on page 16

9. Please move the text box below the reference list.

Move made.

10. Please include a figure title and legend section below the reference list.

Included.
your sincerely